Table 1: November 18, 1998 - Subsystem Status

SS

SS

No. Lead Status Problems
1.0 | Escuadra = Continuing work on the Automated CoastLine
/Cooper detection. (Nguyen)

Continue working with DAAC to checkout update
to DPREP processing scripts. Looking at new date
verify that it is the same. (Cooper)

Working on the SS1 delivery scheduled for Nov.
18th. Currently doing integration tests. This delive]
adds Partial Earth-viewing footprints to the IES,
fixes wrapped lines in the QC reports, adds a ne\
web directory to SS1, fixes errors in flagging of
radiances, and fixes the spaceclamp algorithm tg
throw out data when the instrument scans the Moq
(Cooper, Rodier)

Testing updates to SpaceClamp algorithm to thro
out scans that scanned the Moon. Analysis show
that the Shortwave spaceclamps on RAPS days
noisy. (Anselmo, Spence)

Continuing investigation of the DAA + 15V
anomaly. (Hess, Spence, Weaver)

Worked on getting data into ACCESS database f
the DAA +15V anomaly to more easily access da
during certain events/conditions. (Rodier)
Updating Delivery Memo and Test Plan for the
Instrument Subsystem delivery on Nov. 18th. (Fil
Continue operational support for TRMM and EOS
AM1. (Weaver)
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No.

SS
Lead
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2.0

Chang

Modified all ERBE-like scripts, for on-line and off-

line, to change Sampling Strategy variables SS2
SS2.1,SS2 3t0SS2and SS3 as per Maria’s reqU
(Chang)

Removed ES-8 HDF-EOS verification from PGE
CER2.3P1. ES-8 HDF-EOS verification programs

were modified and moved to test_suites directory.

New makefiles were created. New instructions ar
README file were written for CM SSI&T testing.
(Chang)

Reviewed SS2 Operator’s Manual with the Syste
Engineering Committee. (Chang, Snell)

Started working on F90 HDF read routines that c
be used to read all ERBE-like HDF products.
(Chang)

Continued to work on getting existing Web
applications to run on the new Web server.
Everything is now in place except the ADM graphi
(igloo plots). (Flug)

Investigating the possibility of changing ES-9 HD
from 7 to 155 SDSs. (Snell)

Converted ES-8 HDF-EOS read sample code fron
to Fortran90. (Snell)
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3.0

Chang

Combined with above.
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4.1

Murray

Worked on the transition of the Cloud Web page and
web utilities over to the new web server. Continued

to help team members with the transition of their
pages. Began work on moving the QC Displayer
over to the cwg account and setting up Action Ite
utility for NASA/AS&M/SAIC Cloud Team.
(Brown)

After correcting the default logic for the VIRS
radiances, a lot more bad data appeared in the d
stream. Validated that changes were correctly
implemented. (Sun-Mack)

With the monthly QC results from January 1998,

m

ata

performed a comparison of clear sky reflectance and

observed 0.6um reflectance under daytime/stron
clear conditions. Also performed a comparison of
clear sky predicated T3-T4 and observed T3-T4
under the same conditions. (Sun-Mack)
Worked with Stowe’s group to retrieve a version 4
VIRS file and process through Cloud Retrieval to
generate the AOT off-line product for their study.
(Sun-Mack)

Generated a monthly QC product from which to
generate the 1.6 micron overhead albedo start-u

’

|

D

mean and standard deviation maps for incorporation

into the production code. Used default values (IG}
type-averaged) to fill in areas with no observed
values. Generated production style maps. Integra
the data into the production code. (Murray/Sun-
Mack)

Worked to transfer web pages from Iposun to the
post-firewall configuration. (Sun-Mack/Murray/
Brown)

Began preparations for the Clouds Delivery. Will
deliver after Instrument. (Murray)

Identified and removed the cause of the recently
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developed 'Frozen-Ocean’ problem. (Murray)
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4.1

Murray

Consulted with Alice Fan to determine the source
the problem where the Cloud Code was not
producing the met files associated with the Subs
CloudVis files. Added the optional "Version’
parameter to the code that calls the WriteMeta util
from cereslib. (Murray)

Began analysis of the Version 4 VIRS data.

Processed December 28, 1997 hour 00 and all of

January 5, 1998. Results are still under investigati
(Murray)
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4.2

Murray

Combined with above.

4.3

Murray

Combined with above.

4.4

Miller

Analyzed the reprocessed VIRS with and without
the CERES gain. Provided broad-narrowband
regression statistics. Also, obtained global cloud
properties mean. (Miller)

Downloaded reprocessed VIRS data for January
and 12, 1998 from Goddard DAAC. VIRS
reprocessed through Jan 16 as of Nov 16. (Mille
Developed script for chron jobs in producing daily
binary QC files. Included ability to read chron job
file in binary QC report software. (Dunton, Miller
Continued modification to ERBE-like graphics coq
to display SSF cloud properties. (Dunton, Miller)

5

je

Resolved problems with convolution web page when

new server renamed Iposun. Improved look of
convolution home page. (Dunton)

Updated and tested convolution code to handle n
SSF type definition. (Miller)

Completed changes to Operations Guide, Delive
Memo and Test Plan. (Miller)

Converted VIRS regression code into a module t
pass to inversion upon final NASA coordination.
(Miller)

Found on the web the necessary timing informati
for each MODIS imager pixel need in cloud
retrieval. (Miller)

Continued validation of the TRMM SSF using DX.

(Miller)
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4.5

Nolan

Delivered Subsystems 4.5 and 4.6. software and
Plan to CERES CM on November 13, 1998. This
software and test case use the latest ssf_typdef 3
SSF_ID 113. (Nolan and Franklin)

Completed final draft of the Subsystems 4.5 and
Operator's Manual and submitted to the System

Engineering Committee for review as an examplg
Operator’'s Manual. (Nolan)

Created 24 binary SSFs for 01-05-1998 at the S(

These SSFs used an earlier version of the ADMS.

(Nolan)

Created subsetted SSFs for April 1998 RAPS hou
(Nolan)

Continued to work on the module that reads the S
HDF file into a structure that can be printed by th
ssfread program. (Franklin)

Completed work on modifying the ssf2hdf code tq
create 127 SDSs. A sample SSF HDF file with 12
SDSs can be found on lightning at the following
address: /CERES/inversion/data/scr/
CER_SSF_TRMM-PFM-
VIRS_ValR3_000000.1998010510.127SDSs.
(Birch and Franklin)

Continued work on updating QC software for
Subsystems 4.5 and 4.6. (Nolan and Birch)
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4.6

Nolan

Combined with above.

5.0

Coleman

Continued with modifications for adding the wind
channel output data. (Coleman)
Continued testing and studying the Synoptic SAR
scripts and generators. (Snell)
Modifying Operator’s Manual for 5.0 as per reviey
committee’s suggestions. (Coleman)
Assisting Fred Rose in his quest for determining
cause of a SW albedo (TRMM-SARB) bias.
Uncovered and corrected a minor error in the
distribution of cloud area fraction adjustments, bd
bias remains. (Coleman)
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7.2

Coleman

Combined with above.
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12.0

Coleman

Modified Operator's Manual as per review
suggestions and sent to Maria. (Kizer)
Prepared a package of software for the DAAC fo
use in subsetting the MOA data as per Anand
Inamdar’s request. (Kizer)

Prepared README file for MOA subsetting
software. (Kizer, Coleman)

Preparing a new delivery package for Regrid MOA

Subsystem. This package enables use of backup
NCEP meteorological data. (Kizer)

Expanded testing procedures and test plan for Regrid

MOA Subsystem to include test cases using the
backup meteorological and ozone data sources.
(Kizer, Coleman)

Working with Don Brown and Roberta Dipasquals
in their efforts to regrid some of their data. (Kizel
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7.1

Jimenez

Combined with below.

8.0

Jimenez

Combined with below.

10.0

Jimenez

Continued testing new solar_declination module
time conversion routines. (Jimenez, Raju)
Nichele provided us with new SFCs which we will
begin using to test and validate SS10. (Jimenez
Raju)

Met with Cathy Nguyen to discuss what issues
remain with TISA Averaging, and what the future
may hold. (Jimenez, Raju)

Continued looking at the surface algorithm routines.

(Raju)
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6.0

McKoy

Delivered the corrected routines to convert GMT
Local time to CERESLIib this week. (McKoy)
Made the corrections and/or modifications to the
Operator's Manual that were suggested SEC. Th
new template for the Operator's Manual still neeg
to be applied. (McKoy)

Began updating the TISA Gridding Test Plan and
Delivery Memo to the new format in preparation fg
the next delivery of the TISA Gridding software.
(McKoy)

Met with Henry Flippo to determine how many
TISA Gridding jobs could be run at the SCF
simultaneously. On Lightning, 12 hours of data c:
be run at the same time which is very helpful whe
attempting to run a month of data for validation. Tf
January 1998 data has been run at the SCF and it
been provided the data to TISA Averaging.
Currently validating the results of this run. (McKo
Validating TISA Gridding Jan. 1998 output.
(McKoay, Costulis, Young)
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9.0

McKoy

Combined with above.

11.0

Stassi/
Fan

Developed an IDL program to display GOES-8
hourly raw data as a movie-like image. This progra
finds all available input files which match the data
date entered and displays the images sequential
(Fan)

Modified the PCF ASCII input file generator to
check for matching pairs of GOES-8 image files a|
orbital files for each hour. (Fan)

Ran 199810 GOES-8 through the main and post
processors. The statistics were collected for
streaking analysis. (Fan)

Reviewed draft of GGEO Operator’'s Manual.
(Mitchum, Nolan, Travers, Stassi, Fan)

Started modifying Operator's Manual based on
comments received during review. Also, renamed
dozen GGEO scripts based on comments from
review [ex: run_main.csh -> run_ggeomain.csh]
(Stassi)
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CERESIib
Stassi/ Fan

Wrote up CERESIib Update procedures and started
writing up procedures for making CERESIib DAAC

deliveries. (Stassi)
Wrote a script for automating the CERESIib
validation versions on samantha from lightning.

Wrote a script to automate the tarring of CERESIib

in preparation for DAAC delivery. (Stassi)

Tested the streamlined Toolkit. MTD. Found that @

function called by libPGSTK.a is not included.
Reported this to Toolkit people. (Fan)

CM | Ayers

Tested CERES Subsystems 4.5 & 4.6 (Inversion
and CERESIib, and released them to the DAAC.
(Ayers)

IST | Flug

Moved the TRMM IST Web site to the new Web
server. Converted the plotting program from PV-
Wave to IDL.
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